Author’s response to reviews

Title: Effectiveness of a decision aid for promoting colorectal cancer screening in Spain: a randomized trial

Authors:

Lilisbeth Perestelo-Perez (lilisbethp@gmail.com)
Amado Rivero-Santana (amado.riverosantana@sescs.es)
Alezandra Torres-Castaño (atorrcas@sescs.es)
Vanesa Ramos-García (vanesa.ramosgarcia@sescs.es)
Yolanda Alvarez-Perez (yolanda.alvarezperez@sescs.es)
Nerea Gonzalez-Hernandez (NEREA.GONZALEZHERNANDEZ@osakidetza.eus)
Andrea Buron (ABuron@parcdesalutmar.cat)
Michael Pignone (Michael_Pignone@med.unc.edu)
Pedro Serrano-Aguilar (pseragu@gobiernodecanarias.org)

Version: 4 Date: 27 Dec 2018

Author’s response to reviews:

Santa Cruz de Tenerife, December 27, 2018

Dear Editor:

Please find enclosed our revised manuscript entitled “Effectiveness of a decision aid for promoting colorectal cancer screening in Spain: a randomized trial” (MIDM-D-17-00233R3) by Lilisbeth Perestelo-Perez, Amado Rivero-Santana, Alezandra Torres-Castaño, Vanesa Ramos-Garcia, Yolanda Alvarez-Perez, Nerea Gonzalez-Hernandez, Andrea Buron, Michael Pignone, and Pedro Serrano-Aguilar, for its consideration as an original article in BMC Medical Informatics and Decision Making.
We would like to thank the reviewers and you for your careful revision and thoughtful comments on our paper. All the comments have been carefully taken into consideration for the preparation of our revised. We appreciate the constructive feedback that without doubt has improved our original.

Yours sincerely,

Lilisbeth Perestelo-Perez, MPsych, Ph.D.
Servicio de Evaluación y Planificación
Dirección del Servicio Canario de la Salud
Camino Candelaria, 44. C.S. El Chorrillo
lperperr@gobiernodecanarias.org
lilisbethp@gmail.com
Tel. +34 922 684019 (Ext 241)